Author's response to reviews

Title: Performance of the Pediatric Index of Mortality 2 (PIM-2) in cardiac and mixed Intensive Care Units in a tertiary children's referral hospital in Italy

Authors:

Marta L Ciofi degli Atti (marta.ciofidegliatti@opbg.net)
Marina Cuttini (marina.cuttini@opbg.net)
Lucilla Ravà (lucilla.rava@opbg.net)
Silvia Rinaldi (silvia.rinaldi@opbg.net)
Carla Brusco (carla.brusco@opbg.net)
Paola Cogo (paola.cogo@opbg.net)
Nicola Pirozzi (nicola.pirozzi@opbg.net)
Sergio Picardo (sergio.picardo@opbg.net)
Franco Schiavi (franco.schiavi@opbg.net)
Massimiliano Raponi (massimiliano.raponi@opbg.net)

Version: 5 Date: 10 June 2013

Author's response to reviews: see over
To the Editor
BMC Pediatrics

Dear Editor,

We have revised the manuscript entitled “Performance of the Pediatric Index of Mortality 2 (PIM-2) in cardiac and mixed Intensive Care Units in a tertiary children’s referral hospital in Italy”.

Please find below a point-by-point response to Reviewers’ concerns which have been addressed in the revised manuscript.

Reviewer: Angela S Czaja

Response to Reviewer

The word chi-squared in the methods section is misspelled as "che-square". *Typo has been amended.*

Thanking you in advance for your consideration.

Sincerely,

Marta Luisa Ciofi degli Atti, M.D.
Medical Direction
Bambino Gesù Children’s Hospital
Piazza S. Onofrio, 4
00161 Roma, Italy.
Tel +39 06 68592319
Fax +39 06 68592100
e mail: marta.ciofidegliatti@opbg.net